Author’s response to reviews

Title: Elastography for the Diagnosis of High-Suspicion Thyroid Nodules Based on the 2015 American Thyroid Association Guidelines: A Multicenter Study

Authors:

Hairu Li (lihairu0419@126.com)
Yulan Peng (yulanpeng@163.com)
Yan Wang (Msyanwang1998@126.com)
Hong Ai (hong-ai@163.com)
Xiaodong Zhou (zhouxd@fmmu.edu.cn)
Lichun Yang (yn_ylc@163.com)
Kun Yan (ydbz@sina.com)
Ying Xiao (Laoer182002@aliyun.com)
Lisha Liu (gxwgaoxw@sina.com)
Baoming Luo (bmluo2005@126.com)
Qiang Yong (cfp99@163.com)
Shuzhen Cong (shzhcong@163.com)
Shuangquan Jiang (jsq1982@163.com)
Xin Fu (136897780@qq.com)
Buyun Ma (ws_mby@126.com)
Yi Li (13651873180@163.com)
Xixi Zhang (740530962@qq.com)
Xue Gong (gongxuedier@126.com)
Haitao Chen (736842555@qq.com)
Wenying Liu (liuwy23@sina.com)
Ling Tang (45759019@qq.com)
Author’s response to reviews:

1. If written informed consent for participation was obtained from all participants, please explicitly state this in the "Ethics approval and consent to participate" section.

   The sentence "Written informed consent for participation was obtained from all patients." was added to the manuscript(Page 5, line 26-27;Page 14, line 23).